Author's response to reviews

Title: The direction of research into visual disability and quality of life in glaucoma.

Authors:

   Fiona C Glen (f.c.glen@googlemail.com)
   David P Crabb (david.crabb.1@city.ac.uk)
   David F Garway-Heath (David.Garway-Heath@moorfields.nhs.uk)

Version: 2 Date: 23 March 2011

Author's response to reviews: see over
RE: Glen et al
The direction of research into visual disability and quality of life in glaucoma

Dear Sir/Madam,

Following our recent communication we would be pleased if you could re-consider our manuscript for publication in BMC Ophthalmology as a ‘Correspondence Type 3’ article. In line with suggestions made by the editor we have modified the manuscript accordingly.

Quality of life is an area which is often overlooked in glaucoma research, despite its obvious importance for patient management and overall wellbeing. We believe this study raises awareness of this point, and identifies a need for further studies into the impact of glaucoma on everyday functioning.

I certify on behalf of myself and all co-authors that we have all participated sufficiently in the study design, interpretation of the data and the writing of the manuscript.

We hope you find the paper interesting and relevant to this journal.

Best wishes,

Fiona Glen